Author’s response to reviews

Title: Predictive ability of an early diagnostic guess in patients presenting with chest pain; a longitudinal descriptive study.

Authors:

François Verdon (Francois.verdon@mcnet.ch)
Michel Junod (junod.nicod@bluewin.ch)
Lilli Herzig (lilli.herzig@hin.ch)
Paul Vaucher (paul.vaucher@hospvd.ch)
Bernard Burnand (bernard.burnand@chuv.ch)
Thomas Bischoff (thomas.bischoff@hospvd.ch)
Alain Pécoud (alain.pecoud@hospvd.ch)
Bernard Favrat (bernard.favrat@hospvd.ch)

Version: 5 Date: 22 January 2010

Author’s response to reviews:

Dear Editor,

We have revised the abstract to structure it as requested by the BMC guidelines. The entire manuscript was sent and edited by SanFrancisco Edit. This has brought many syntax changes but none of which modified the meaning or the content of the article.

The article has been formatted following instructions for publication. We hope this version brings you entire satisfaction.

Regards,

Paul Vaucher